

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2021JG006539

Key Points:

- N addition increased soil N₂O emission in the valley across the 3 years
- High N addition stimulated N₂O emission on the slope at the early stage
- Moderate N stimulated greater N₂O emission in the valley than on the slope

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. Li, dejunli@isa.ac.cn

Citation:

Duan, P., Wang, D., Xiao, K., Zheng, L., Chen, H., Wang, K., & Li, D. (2022). Responses of soil nitrous oxide emission to nitrogen addition at two topographic positions of a subtropical forest. *Journal* of Geophysical Research: Biogeosciences, 127, e2021JG006539. https://doi. org/10.1029/2021JG006539

Received 23 JUL 2021 Accepted 1 FEB 2022

Author Contributions:

Conceptualization: Pengpeng Duan, Dejun Li Data curation: Liang Zheng Funding acquisition: Pengpeng Duan, Daobo Wang, Dejun Li Methodology: Pengpeng Duan, Liang Zheng Resources: Hao Chen Software: Pengpeng Duan Supervision: Kongcao Xiao, Liang Zheng, Hao Chen, Kelin Wang Visualization: Dejun Li Writing - original draft: Pengpeng Duan Writing - review & editing: Daobo Wang, Kongcao Xiao, Kelin Wang, Dejun Li

© 2022. American Geophysical Union. All Rights Reserved.

Responses of Soil Nitrous Oxide Emission to Nitrogen Addition at Two Topographic Positions of a Subtropical Forest

Pengpeng Duan^{1,2,3} ^(D), Daobo Wang⁴, Kongcao Xiao^{1,2,3}, Liang Zheng⁵, Hao Chen⁶, Kelin Wang^{1,2,3} ^(D), and Dejun Li^{1,2,3} ^(D)

¹Key Laboratory of Agro–Ecological Processes in Subtropical Region, Institute of Subtropical Agriculture, Chinese Academy of Sciences, Changsha, China, ²Guangxi Industrial Technology Research Institute for Karst Rocky Desertification Control, Nanning, China, ³Guangxi Key Laboratory of Karst Ecological Processes and Services, Huanjiang Observation and Research Station for Karst Ecosystems, Institute of Subtropical Agriculture, Chinese Academy of Sciences, Huanjiang, China, ⁴College of Biology and Pharmacy, Yulin Normal University, Yulin, China, ⁵Urban Construction College, Shaoyang University, Shaoyang, China, ⁶State Key Laboratory of Biocontrol, School of Ecology, Sun Yat-sen University, Guangzhou, China

Abstract Topography can influence nitrous oxide (N₂O) emission via its influences on soil nutrient availability, moisture, and microbial communities. Nevertheless, it is still unclear whether topography modulates the responses of soil N₂O emissions to elevated N deposition. Here the N addition experiment was conducted in the valley and on the slope of a subtropical karst forest in southwest China. Nitrogen was applied as NH₄NO₃ in two levels, that is, 50 (moderate N) and 100 (high N) kg N ha⁻¹ yr⁻¹ with no N addition plots as the control. Nitrogen addition consistently increased N₂O emission in the valley, but only high N addition significantly increased N₂O emission on the slope in 2017. The cumulative N₂O fluxes across the 3 years were 1.16 ± 0.24 kg N ha⁻¹ in the valley and 1.50 ± 0.06 kg N ha⁻¹ on the slope under the control. Nitrogen addition stimulated N₂O emission by 88.7%–113.3% in the valley due to increased ammonium, nitrate and dissolved organic carbon availabilities and ammonia-oxidizing bacteria (AOB) *amoA* abundance. High N addition stimulated N₂O emission by 84.3% on the slope owing to increased nitrate and carbon availabilities, AOB *amoA*, and *nirK* abundances. The stimulation of N₂O emission by moderate N addition was more pronounced in the valley than on the slope largely owing to the lower N status in the valley. This work highlights the importance of N status in regulating the responses of soil N₂O emissions to elevated N deposition.

Plain Language Summary Atmospheric N deposition is identified as one of the strongest driving factors responsible for the increase of forest soil N_2O emission, and topography can influence N_2O emission via its influences on soil nutrient availability, moisture, and microbial communities. However, there is uncertainty about whether topography modulates the responses of soil N_2O emissions to elevated N deposition. We investigated soil N_2O emission and related microbial functional gene abundances at two topographic positions under three N addition levels in a subtropical region of China. We find that the N addition consistently increased N_2O emission in the valley, but only stimulated N_2O emission at the early stage on the slope. Stimulation of soil N_2O emission by moderate N addition was more pronounced in the valley than on the slope. Our findings highlight the importance of N status in regulating the response of soil N_2O emissions to elevated N deposition.

1. Introduction

Nitrous oxide (N₂O) is a potent greenhouse gas in the troposphere with an atmospheric lifetime of ~116 years and contributes to ozone depletion in the stratosphere (Oikawa et al., 2015; Ravishankara et al., 2009). Forest soils are a significant source of atmospheric N₂O with a mean flux of 3.62 ± 0.16 Tg N yr⁻¹ (1 Tg = 10^{12} g) and an increment rate of about 9.9×10^6 g N yr⁻¹ from 1992 to 2015 (Zhang et al., 2019). Atmospheric nitrogen (N) deposition is identified as one of the strongest driving factors responsible for the increase of soil N₂O emissions (H. Tian et al., 2019). However, the contribution of elevated N deposition to the increase of soil N₂O emission from forests is poorly constrained with a range from 92% to 739%, largely due to uncertainties in the understanding of mechanisms responsible for soil N₂O emission under elevated N deposition (Deng et al., 2020; Liu & Greaver, 2009). Considering that global N deposition is projected to be doubled in 2050 relative to the level of 100–115 Tg yr⁻¹ in 2000 (Galloway et al., 2004), it is hence imperative to better understand the mechanisms underlying the impacts of elevated atmospheric N deposition on soil N₂O emission.



Multiple mechanisms, including alteration of N and phosphorus (P) status, an increase of ammonium (NH_4^+) and nitrate (NO_3^-) availability, and soil acidification, have been proposed to explain the responses of soil N_2O emission to elevated N deposition (Deng et al., 2020; Hall & Matson, 1999; Koehler et al., 2009; Martinson et al., 2013; Redding et al., 2016; Rowlings et al., 2012). However, the results from individual studies in the subtropical/tropical region are inconsistent. For instance, Han et al. (2019) found that experimental N addition (N addition hereafter) increased N₂O emission under N poor conditions via stimulating the abundances of nitrifiers and denitrifiers, but decreased N₂O emission in N rich conditions by depressing the abundances of denitrifiers in a subtropical forest. M. H. Zheng et al. (2016) observed significant increases in soil N₂O emission under N addition only increased N₂O fluxes during the periods with high soil moisture content. Corre et al. (2014) also reported that the response of soil N₂O emissions from tropical forests to N addition was more pronounced in wet years than in dry years. It seems that initial N status and moisture contents affect the response of soil N₂O emission to N addition in the subtropical/tropical forests.

Soil N₂O is mainly produced in nitrification and denitrification processes (Butterbach-Bahl et al., 2013; Duan, Zhou, et al., 2019). During nitrification, N₂O production is regulated by *amoA* gene encoding the ammonia monooxygenase enzyme of ammonia-oxidizing archaea (AOA) and ammonia oxidizing bacteria (AOB; Prosser et al. [2020]). During denitrification, a key step, in which nitrite is reduced to nitric oxide (NO), is catalyzed by nitrite reductases encoded by *nirK* and/or *nirS* genes (Clark et al., 2012). Another important denitrification step for N₂O emission is the reduction of N₂O to N₂, which is catalyzed by nitrous oxide reductase encoded by the *nosZI* and/or *nosZII* genes, representing the only known microbial sink for N₂O (Hallin et al., 2018). Since soil N₂O emission is almost entirely controlled by microbial activities, the quantification and characterization of specific functional genes involved in nitrification and denitrification pathways help predict N₂O emission (Hu et al., 2015; Levy-Booth et al., 2014). For example, high N₂O emission from arctic peatlands is primarily fueled by nitrification mediated by AOA. Higher *nirK* than *nosZ* abundance in a subtropical forest induced more N₂O production than N₂O reduction, resulting in a high contribution of denitrification to N₂O emission (Han et al., 2018). Although microbial regulation of N₂O emission has been widely investigated (Banerjee et al., 2016; Duan, Zhou, et al., 2019; Srikanthasamy et al., 2018), limited studies have been conducted in terms of microbial regulation of soil N₂O emission in responses to N addition.

Topography controls soil N status, soil N transformation rates, soil organic carbon level, soil microbial community composition, and abundance, soil moisture, etc., which may subsequently affect soil N_2O emission (Arias-Navarro et al., 2017; Enanga et al., 2016; Stewart et al., 2014). For example, it has been reported that the lower slope positions are hot spots of N_2O emission due to high denitrification (Stewart et al., 2014; Yu et al., 2019). Topography can also influence N availability and the fate of atmospherically deposited N via its influences on soil nutrient availability, moisture, and microbial communities (Enanga et al., 2017). Since N status likely affects the response of soil N_2O emission to N addition as aforementioned, it is possible that the responses of soil N_2O emission to N addition would vary among topographic positions.

In the current study, a field N manipulation experiment was conducted at two topographic positions, that is, in the valley and on the slope of a subtropical karst forest with calcareous soil in southwest China (Wang et al., 2019). Calcareous soils cover more than 30% of the Earth's surface (Bertrand et al., 2007). These soils are usually characterized by high Ca and Mg contents, and would therefore exhibit high soil buffering capacity to N fertilization (Wang et al., 2019; L. Zheng et al., 2020). Moreover, elevated N addition can promote nitrification in calcareous soil without acidification (Hao et al., 2020). We, therefore, hypothesized that N addition would increase soil N₂O emission at both topographic positions by increasing the abundance of nitrifiers because of the rapid utilization of N substrate from N addition (**H1**). In a previous study, we reported that asymbiotic N₂ fixation in soil was inhibited by N addition in the valley but not on the slope due to the difference in soil N status, which is higher on the slope than in the valley as indicated by NO₃⁻ leaching (Wang et al., 2009), we hypothesized that topography would modulate the responses of soil N₂O emission to N addition (**H2**). Accordingly, the major questions we aimed to address include: (a) are the responses of soil N₂O emission to N addition in the valley and on the slope controlled by the same set of variables and (b) how does topography modulates the responses of soil N₂O emission to N addition?



2. Materials and Methods

2.1. Site Description and Experimental Design

The N addition experiment was conducted in a subtropical karst forest in Mulun National Nature Reserve of Huanjiang County, southwest China (Wang et al., 2019). This forest was developed from extensively disturbed (biomass harvesting and grazing) shrubland 35 years ago. The area has a humid subtropical climate with mean annual precipitation and air temperature being 1,389 mm and 19°C, respectively. *Cryptocar yachinensis* (Hance) Hemsl., *Cinnamomum saxatile* H.W.Li, *Koelreuteria minor* Hemsl., *Pittosporum tobira* (Thunb.) Ait., *Bridelia tomentosa* Bl., and *Murraya exotica* L. Mant are the dominant tree species. The soil in the karst forest is classified as leptosols (limestone soil) according to the World Reference Base classification system. Soil depth in the valley and on the slope varies from 0.0 to 0.8 m and 0.0–0.3 m, respectively. Atmospheric N deposition rate was approximately 37 kg N ha⁻¹ yr⁻¹ according to Zhu et al. (2015).

The experiment was carried out in the valley and on the slope, respectively (Wang et al., 2019; Figure S1). For each topographic position, a randomized complete block design was adopted with three blocks and three N addition levels, that is, control (N0, 0 kg N ha⁻¹ yr⁻¹), moderate N addition (N50, 50 kg N ha⁻¹ yr⁻¹) and high N addition (N100, 100 kg N ha⁻¹ yr⁻¹). Therefore, there are nine plots (10×10 m each) at each topographic position. Each plot was surrounded by a 10 m wide buffering zone to avoid potential influence from each other (Figure S1). Nitrogen was applied in the form of NH₄NO₃ at the beginning of each month since April 2016. On each N addition event, NH₄NO₃ was dissolved in 10 L of water (equivalent to 1.2 mm precipitation per year) and sprayed with a back–pack sprayer. The control plots received an equal amount of water.

2.2. Determination of N₂O Fluxes

At each plot, two static PVC chambers were deployed for the determination of N₂O fluxes twice a month from November 2016 to October 2019. Each chamber consisted of an anchor ring (25 cm in inner diameter and 10 cm in height, inserted 3 cm into the soil) and a removable cover chamber (25 cm in inner diameter and 35 cm in height). Sampling was conducted between 8:00 and 11:00 in the morning at the mid and end of each month. It should be noted that the sampling scheme was chosen to avoid measuring N₂O fluxes immediately after N addition when transitory peaks were expected. Gas samples were taken at 0, 10, 20, and 30 min after chamber closure using a 60 mL plastic syringe and transferred immediately into a 12-ml pre-evacuated glass vial (Labco Exetainer, Labco Limited, UK). N₂O was measured using a gas chromatograph equipped with an FID detector (Agilent GC 7890A, Agilent, USA) within 1 week after collection. We considered the linearity and nonlinearity of flux calculation when designing the sampling (Levy et al., 2011). First, we conducted a preliminary experiment to determine the time range in which N₂O concentration in the chamber increased linearly. We found that N₂O concentration in the chamber increased linearly within 60 min. Second, we assessed each group of data to judge whether N₂O concentration increased linearly within 30 min when calculating N₂O flux. We found that linearity was applicable to all the data sets, so the N₂O fluxes were calculated from the linear increase of N₂O concentration in the chamber versus time and were adjusted for the field-measured air temperature and atmospheric pressure (Koehler et al., 2009). Cumulative N₂O emissions were calculated assuming constant flux rates between two sampling dates (Duan, Zhang, et al., 2019).

2.3. Auxiliary Analysis

Soil temperature and volumetric moisture at 5 cm depth were simultaneously measured using a digital thermometer (JM 624, Jinming Instrument Co. Ltd., Tianjin, China) and a portable soil moisture meter (Mode TZS–1K, Zhejiang Top Instrument Corporation Ltd., China), respectively, along with gas sample collection. Volumetric soil moisture content was further converted to a percentage of water-filled pore space (%WFPS) based on soil bulk density and volumetric water content.

Soil samples to a depth of 0–10 cm were collected once every 3 months after litter removal. At each plot, three soil samples were collected using a soil corer of 2 cm in diameter and mixed thoroughly to form a composite soil sample. The soils were sieved through a 2–mm mesh sieve and were separated into two parts. Samples were extracted within 24 hr after field sampling. Soil NH_4^+ and NO_3^- were extracted with 2 M KCl (w/v of 1/5) and analyzed by an auto-analyzer (FIAstar 5000, FOSS, Sweden). DOC was extracted with 0.5 M K_2SO_4 (w/v of 1/5)



and analyzed by the wet–oxidation method (Carter & Gregorich, 2007). Soil NH_4^+ , NO_3^- , and DOC intensities (mg N or C g⁻¹ d⁻¹) were calculated using trapezoidal integration of the content versus time. Nitrogen or C intensity is similar to a time-weighted average of N or C content but includes the duration of exposure, and may reflect longer-term impacts of C or N availability on microbial processes (Maharjan & Venterea, 2013).

Soil samples were collected from each plot on 24 September 2019 for DNA extraction. Soil DNA was extracted using a FastDNA Spin kit for soil (MoBio Laboratories, Carlsbad, CA, USA) according to the manufacturer's instructions. The purified DNA was stored at -20° C until further analysis. Quantitative PCR (qPCR) was used to amplify the targets of N₂O–producers (AOB and archaea (AOA) ammonia monooxygenase genes (*amoA*), bacterial nitrite reductase genes (*nirS* and *nirK*) and fungal nitrite reductase gene (fungal *nirK*)) and N₂O–reducers (nitrous oxide reductase genes (*nosZI* and *nosZII*)) using a Bio–Rad CFX96 Real–Time PCR Detection System (Bio–Rad Laboratories, Inc.). Briefly, PCR reactions were performed in a 20 µl solution including 10.0 µl of SYBR Premix Ex Taq (TaKaRa Biotechnology Co. Ltd., Dalian, China), 2 µl of soil DNA, and 0.25 µM of each primer. The thermal conditions of the PCR reactions were similar to those previously reported for the functional genes (Harter et al., 2014; Jones et al., 2013; Long et al., 2015). The amplification resulted in single peaks with efficiencies of 90.9%–101.2% and R² values of 0.991–0.997. Each qPCR run included soil DNA samples, corresponding standards, and no-template control reactions. Melting curve analysis was performed at the end of each run to confirm the reaction specificity. The gene-specific primers are presented in Table S1.

2.4. Statistical Analysis

In order to assess whether the responses of N_2O fluxes to the same level of N addition differed between the two topographic positions, N-induced change of soil N_2O emission was calculated by Equation 1:

$$Change(\%) = \frac{X_t - X_c}{X_c} \times 100 \tag{1}$$

where X_t and X_c represent cumulative N_2O emission from a N addition plot and the control, respectively, in the same block. One way analysis of variance (ANOVA) was used to examine the effects of N addition on cumulative N_2O emission, intensities of NH_4^+ , NO_3^- , DOC, and DOC: NO_3^- ratio, and functional gene abundances for each topographic position. Independent samples *t*-test was used to examine whether the difference is significant between the two topographic positions for the same N addition treatment. Regression analysis was used to identify significant correlations of cumulative N_2O emission with substrate concentrations or gene abundances. The effect was considered significant at *p* < 0.05 level. Linear mixed effect models (LMEM) were used to examine the effects of N addition, topography, sampling date, and their interactions on soil N_2O flux and N-induced change of N_2O flux and with a plot nested in time as a random effect. The above analyses were performed in SPSS 25.0 (IBM Co., Armonk, NY, USA).

3. Results

3.1. Soil Temperature, WFPS, and N₂O Emission

Soil temperature and water content varied from 4 to 38°C (Figures 1a and 1b) and 15%–90% WFPS (Figures 1c and 1d), respectively, over the measurement period. The averaged N₂O fluxes for the control were $4.43 \pm 5.49 \ \mu g$ N m⁻² h⁻¹ in the valley and $6.12 \pm 5.89 \ \mu g$ N m⁻² h⁻¹ on the slope (Figures 1e and 1f). Soil N₂O emission was significantly influenced by N addition in 2017, 2018, and 2019, and topography in 2018 (Table S3). Although there was no significant effect of topography on N₂O flux in 2017 and 2019, the significant interactive effect of topography × N addition × sampling date and topography × N addition was found in 2017 and 2019, respectively (Table S3).

Cumulative N₂O emission was significantly influenced by N addition (p < 0.05; Figure 2). The cumulative N₂O emissions in 2017, 2018, and 2019 were 0.59 ± 0.07 kg N ha⁻¹, 0.35 ± 0.03 kg N ha⁻¹, and 0.22 ± 0.18 kg N ha⁻¹ in the valley, and were 0.46 ± 0.08 kg N ha⁻¹, 0.56 ± 0.14 kg N ha⁻¹, and 0.49 ± 0.13 kg N ha⁻¹ on the slope, respectively, under the control (Figures 2a–2c). Moderate N addition significantly (p < 0.05) increased cumulative N₂O emission by 94.68% in 2018% and 251.17% in 2019 in the valley. High N addition significantly (p < 0.05) increased cumulative N₂O emission by 53.31% in 2017%, 124.30% in 2018%, 252.92% in 2019 in





Figure 1. Temporal variations of soil temperature (a–b), water-filled pore space (c–d), and N₂O fluxes (g–h) in the valley and on the slope, respectively. N0, N50, and N100 indicate control (0 kg N ha⁻¹ yr⁻¹), moderate N (50 kg N ha⁻¹ yr⁻¹), and high N (100 kg N ha⁻¹ yr⁻¹) treatments, respectively. Values are presented as means with standard deviations (n = 3).

the valley, and by 117.54% in 2017 on the slope. Cumulative N₂O emissions for the control were 0.93 ± 0.07 kg N ha⁻¹ and 1.02 ± 0.15 kg N ha⁻¹, respectively, in the valley and on the slope across 2017 and 2018, and were 1.16 ± 0.24 kg N ha⁻¹ and 1.50 ± 0.06 kg N ha⁻¹, respectively, in the valley and on the slope across the 3 years (Figures 2d and 2e). Across the 3 years, moderate and high N addition significantly increased soil N₂O emission by 88.7%–113.3% in the valley, and high N addition significantly increased soil N₂O emission by 88.7%–113.3% in the valley, and high N addition significantly increased soil N₂O emission by 84.27% on the slope. Cumulative N₂O emission across the 3 years was higher in the valley than on the slope only under N50 (p < 0.05; Figure 2e). There was a significant effect of topography on N–induced change of N₂O fluxes across 2017 and 2018, and there were significant interactive effects of topography × sampling date on N–induced change of N₂O emission across the 3 years was significantly, N–induced change of cumulative N₂O emission across the 3). Accordingly, N–induced change of cumulative N₂O emission across the 3 years was significantly in the valley than on the slope under N50 (Figure 2f).

3.2. Soil Abiotic and Biotic Variables

High N addition significantly increased the intensities of soil NH_4^+ and NO_3^- across the 3 years in the valley (Figures 3a–3j). High N addition significantly increased soil DOC intensity in 2019 in the valley (Figure 3m). High N addition significantly decreased soil DOC: NO_3^- ratio across 2017 and 2018 in the valley (Figures 3p and 3q). On the slope, high N addition significantly increased soil NH_4^+ and NO_3^- intensities (Figures 3a and 3f), but decreased soil DOC: NO_3^- ratio in 2017 (Figure 3p). Compared to the control, N addition significantly increased the abundances of AOB *amoA* (Figure 4b) and *nirK* (Figure 4d) on the slope, but decreased (p < 0.05) AOA:AOB *amoA* ratio in the valley (Figure 4c).





Figure 2. Impacts of N addition on cumulative N₂O emission in 2017, 2018, 2019, 2018 + 2019, and across 3 years (a–e), and N–induced change of cumulative N₂O emission (f) in the valley and on the slope. Different letters denote significant differences at p < 0.05 among N addition treatments in the valley or on the slope. The asterisk (*) indicates significant difference at p < 0.05 between the valley and slope for the same N addition level. Values are presented as means with standard deviations (n = 3).

3.3. Relationships Between Cumulative N₂O Emission and Soil Properties

Cumulative N₂O emission across the 3 years was positively (p < 0.05) correlated with the soil NH₄⁺, NO₃⁻, DOC intensities, and AOB *amoA* abundance (Figures 5a–5c, e), but negatively (p < 0.05) correlated with the AOA:AOB *amoA* and DOC:NO₃⁻ ratios in the valley (Figures 5d and 5g). There was a significant positive (p < 0.05) correlation of cumulative N₂O emission with the NO₃⁻ intensity, AOB *amoA* or *nirK* abundance (Figures 5b, 5f and 5h), and negative (p < 0.05) correlation with the DOC:NO₃⁻ ratio (Figure 5d) on the slope.

4. Discussion

4.1. Soil N₂O Emission in the Control Plots

The annual N₂O emissions $(0.39 \pm 0.08-0.50 \pm 0.02 \text{ kg N ha}^{-1} \text{ yr}^{-1})$ in the control plots of the current study are comparable to some previously reported values $(0.39-0.52 \text{ kg N ha}^{-1} \text{ yr}^{-1})$ observed in some subtropical/tropical forests (Fan et al., 2017; Rowlings et al., 2012; Soper et al., 2018), but are much lower than those from other subtropical/tropical forests $(0.70-3.50 \text{ kg N ha}^{-1} \text{ yr}^{-1};$ Han et al., 2019; Wieder et al., 2011; Xie et al., 2018; M. H. Zheng et al., 2016). Several potential mechanisms are responsible for the relatively low N₂O emission in the current study. First, soil water content, with an average of 61.4% WFPS, was high in most of the sampling period (Figure 1). High WFPS may hinder the diffusion of N₂O from the soil profile to the atmosphere (Butterbach-Bahl et al., 2013). Second, complete denitrification (from N₂O to N₂) may dominate the denitrification processes in the forests of the current study, since soil pH (Table S2) is within the suitable range (7.1–7.5) benefiting complete denitrification (Jan-Michael et al., 2018), which ultimately decreases soil N₂O production.

Our results show that there was no significant difference in N_2O emission between the two topographic positions (Figure 2). This is inconsistent with some previous studies which found that topography affects soil N_2O fluxes in forests (Almaraz et al., 2019; Fang et al., 2009; Yu et al., 2021). In these studies, soil moisture (Fang et al., 2009;





Figure 3. Impacts of N addition on the intensities of soil NH_4^+ (a–e), NO_3^- (f–j), DOC (k–o), and DOC: NO_3^- ratio (p–t) in the valley and on the slope. The asterisk (*) indicates significant difference at p < 0.05 between the valley and slope for the same N addition level. Values are presented as means with standard deviations (n = 3).



Figure 4. Impacts of N addition on ammonia oxidizers (a–c) and denitrifiers (d–i) gene abundances in the valley and on the slope, respectively. Different letters denote significant differences at p < 0.05 among N addition treatments in the valley or on the slope. Values are presented as means with standard deviations (n = 3).





Figure 5. Relationships of cumulative N₂O emission with soil substrate intensities (a–d), and functional genes abundances (e–m) in the valley and on the slope. The asterisks * and ** represent significance levels at p < 0.05 and p < 0.01, respectively.

Yu et al., 2021) and substrate availability (Almaraz et al., 2019) have been used to explain the difference in N_2O emission among topographic positions. However, levels of these variables were similar between the two topographic positions in the current study (Figure 3, Table S2).

4.2. Factors Responsible for the Stimulation of Soil $\rm N_2O$ Emission by Nitrogen Addition Are Topography-Dependent

We find that N addition consistently stimulated soil N_2O emission in the valley across the 3 years. Some studies show that the stimulation of N_2O emission is caused by N addition–induced soil acidification (Tian & Niu, 2015), since acidification promotes AOA–driven nitrification (Shi et al., 2018), reduces the consumption of N_2O by denitrification (Koehler et al., 2009) and stimulates fungal driven denitrification (Chen et al., 2015). However, soil pH was not significantly altered by N addition in the current study (Table S2), so the stimulation of soil N_2O emission by N addition should not be caused by soil acidification. An increase in soil N availability has often been used to explain the enhancement of N_2O emission under N addition (Redding et al., 2016). In the current study, two more reasonable mechanisms underlying the increase in soil N_2O emission in responses to N addition are proposed. First, N addition may promote N_2O emission via its positive impacts on soil N availability. As indicated by soil NH_4^+ and NO_3^- intensities, soil N availability was increased by N addition in the valley (Figure 3). Soil



nitrification in the karst forest is fast (Li et al., 2017, 2018), so the deposited NH_4^+ would be rapidly transformed to NO_3^- upon entering into the soil. Therefore, soil NH_4^+ and NO_3^- intensities were found to be simulated by N addition (Figure 3). Enhanced soil NH_4^+ and NO_3^- availabilities resulted in high nitrification and denitrification (Butterbach-Bahl et al., 2013). In the current study, both soil NH_4^+ and NO_3^- intensities had direct positive effects on N_2O emission in the valley (Figure 5, S4), supporting the critical role of N availability in promoting N_2O emission under N addition. Second, N_2O emission may also be promoted by N addition via increased DOC availability. Previous studies reported that high DOC concentrations are indicative of high C availability for denitrification and N_2O emissions (Levy-Booth et al., 2014; Yin et al., 2017). We found that soil NO_3^- and DOC intensities had a positive correlation with the N_2O emission in the valley in 2019 (Figure S4 j, k), indicating that N_2O emission should be more closely related to the denitrification process this year.

In contrast to the valley, N addition only stimulated soil N₂O emission at the early stage on the slope in the current study. This is consistent with the findings of some previous studies (L. Song et al., 2017; X. Song et al., 2020). Similar to the valley, increased N₂O emission by N addition may be attributed to stimulated N availability (Figure 3, S4). Additionally, decreased DOC:NO₃⁻ ratio as a result of limiting DOC availability was reported to inhibit N₂O reduction so N₂O emission increased (Assémien et al., 2019; Grave et al., 2018). Therefore, a decrease in the soil DOC:NO₃⁻ ratio was likely another mechanism underlying the increased N₂O emission on the slope in 2017 in the current study. As aforementioned, the N status and soil NO₃⁻ leaching were higher on the slope after 2017. Due to the high soil nitrification in the karst forest, more deposited NH₄⁺ would be transformed into NO₃⁻ and more input N would be leached on the slope than that in the valley (Wang et al., 2019). This explained why soil NH₄⁺ and NO₃⁻ intensities, and N₂O emission as well were not significantly altered on the slope after 2017.

The stimulation of soil N₂O emission ultimately results from the increase in the activities of nitrifiers and denitrifiers due to the enhanced N substrates under N addition (Tang et al., 2016). AOB is favored by relatively high NH₄⁺ supply (Prosser et al., 2020). In the current study, AOB *amoA* abundance was positively and significantly related to N₂O emission, supporting that AOB was mainly responsible for N₂O production at both topographic positions. AOB has been found to survive under anoxic conditions and contain the *nirK* gene, which is involved in nitrifier denitrification (Wrage-Mönnig et al., 2018). In the current study, *nirK* abundance was positively correlated with AOB *amoA* abundance on the slope (Figure S5), implying that the observed *nirK* gene mainly belonged to AOB. Moreover, positive relationships were observed between N₂O emission and AOB *amoA* or *nirK* abundance (Figures 4e and 4g), indicating the possible contribution of nitrifier denitrification to N₂O emission. High NO₃⁻ accumulation in soil often increases soil N₂O emission via stimulating nitrite reductase rather than reducing N₂O reductase (Giles et al., 2012). Thus, we propose that the increased N₂O emission on the slope in 2017 may be attributed to NO₃⁻ accumulation which accelerates the AOB-driven denitrification. Nitrifier denitrification is prevalent in soils with high ammonia oxidization rates (Butterbach-Bahl et al., 2013; Duan, Zhang, et al., 2019), but this needs to be further confirmed by using double isotopes method.

4.3. Higher Stimulation of Soil N₂O Emission by Nitrogen Addition in the Valley Than on the Slope

Our results show that N addition–induced stimulation of N₂O emission may be modulated by topography with higher stimulation in the valley than on the slope under N50. The topography dependent responses of N₂O emission to N addition are likely caused by soil N status. There is evidence that N addition increased N availability more pronouncedly in N limited than in N rich ecosystems so N addition would lead to greater stimulation of soil N₂O emission in more N limited ecosystems (Fu et al., 2015). According to a meta-analysis, N addition significantly increases soil N₂O emissions in N limited forests but has no significant effect in N rich forests (Deng et al., 2020). With regard to the current study, soil N status is higher on the slope than in the valley as indicated by soil NO₃⁻ leaching, which was higher on the slope (2.14 ± 0.06 kg N ha⁻¹ year⁻¹) than in the valley (1.91 ± 0.04 kg N ha⁻¹ year⁻¹) according to the observation in a neighboring forest (Wang et al., 2019). The soil N:P ratio is often used as an indicator of soil N or P status with a greater ratio implying a higher N status (Tian et al., 2010). The higher soil N:P ratio on the slope than in the valley for the control (Table S2) hence corroborates a higher soil N status on the slope. Due to the lower N status, soil NH₄⁺ intensity was enhanced more pronouncedly in the valley than on the slope under N50 (Figures 3b, 3d and 3e). Nevertheless, functional gene abundances were not significantly different between the valley and slope (Figure S5b, c). Accordingly, the greater stimulation



of soil N_2O emission in the valley should be primarily ascribed to the dynamics of soil N availability under N50. In contrast, high N addition could have increased soil N status in the valley to a level comparable to that on the slope. Due to the relatively high N status, soil N availability was insensitive to N addition both in the valley and on the slope under N100. Subsequently, the degrees of soil N_2O emission responses to high N addition were comparable at the two topographic positions.

5. Conclusions

Our results show that N addition consistently increased N_2O emission in the valley, but only stimulated N_2O emission at the early stage on the slope. Additionally, we find that the stimulation of soil N_2O emission by moderate N addition was more pronounced in the valley than on the slope mostly due to the lower N status in the valley. However, we are not sure if the patterns observed in the current study are applicable to other climate conditions. Since the difference in soil N status among topographic positions is common (Arias-Navarro et al., 2017; Enanga et al., 2016; Stewart et al., 2014), we guess that the topography dependent responses of soil N_2O emission to N addition should be widespread. Our findings highlight that topography should be included in Earth system models in order to better predict the responses of soil N_2O emission to atmospheric N deposition.

Data Availability Statement

The data presented in this work can be found in Dataset S1, and can also be accessed at Science Data Bank (http://www.doi.org/10.11922/sciencedb.01328).

References

- Almaraz, M., Groffman, P. M., & Porder, S. (2019). Effects of changes in nitrogen availability on nitrogen gas emissions in a tropical forest during a drought. Journal of Geophysical Research: Biogeosciences, 124(9), 2917–2926. https://doi.org/10.1029/2018JG004851
- Arias-Navarro, C., Díaz-Pinés, E., Klatt, S., Brandt, P., Rufino, M. C., Butterbach-Bahl, K., & Verchot, L. V. (2017). Spatial variability of soil N₂O and CO₂ fluxes in different topographic positions in a tropical montane forest in Kenya. *Journal of Geophysical Research: Biogeo-sciences*, 122(3), 514–527. https://doi.org/10.1002/2016JG003667
- Assémien, F. L., Cantarel, A. A. M., Florio, A., Lerondelle, C., Pommier, T., Gonnety, J. T., & Le Roux, X. (2019). Different groups of nitrite-reducers and N₂O-reducers have distinct ecological niches and functional roles in West African cultivated soils. *Soil Biology and Biochemistry*, 129, 39–47. https://doi.org/10.1016/j.soilbio.2018.11.003
- Banerjee, S., Helgason, B., Wang, L., Winsley, T., Ferrari, B. C., & Siciliano, S. D. (2016). Legacy effects of soil moisture on microbial community structure and N₂O emissions. *Soil Biology and Biochemistry*, 95, 40–50. https://doi.org/10.1016/j.soilbio.2015.12.004
- Bertrand, I., Delfosse, O., & Mary, B. (2007). Carbon and nitrogen mineralization in acidic, limed and calcareous agricultural soils: Apparent and actual effects. Soil Biology and Biochemistry, 39(1), 276–288. https://doi.org/10.1016/j.soilbio.2006.07.016
- Butterbach-Bahl, K., Baggs, E. M., Dannenmann, M., Kiese, R., & Zechmeisterboltenstern, S. (2013). Nitrous oxide emissions from soils: How well do we understand the processes and their controls? *Philosophical Transactions of the Royal Society B: Biological Sciences*, 368(1621), 91–97. https://doi.org/10.1098/rstb.2013.0122
- Carter, M. R., & Gregorich, E. G. (2007). Soil sampling and methods of analysis (2nd ed.,). Taylor & Francis Group.
- Chen, H., Mothapo, N. V., & Shi, W. (2015). Soil moisture and pH control relative contributions of fungi and bacteria to N₂O production. *Microbial Ecology*, 69(1), 180–191. https://doi.org/10.1007/s00248-014-0488-0
- Chen, H., Tang, J., Sun, X., Ma, K., Chen, H., & Li, D. (2021). Topography modulates effects of nitrogen deposition on microbial resource limitation in a nitrogen-saturated subtropical forest. *Forest Ecosystems*, 8(1), 68. https://doi.org/10.1186/s40663-021-00341-9
- Clark, I. M., Buchkina, N., Jhurreea, D., Goulding, K. W., & Hirsch, P. R. (2012). Impacts of nitrogen application rates on the activity and diversity of denitrifying bacteria in the Broadbalk Wheat Experiment. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 367(1593), 1235–1244. https://doi.org/10.1098/rstb.2011.0314
- Corre, M. D., Sueta, J. P., & Veldkamp, E. (2014). Nitrogen-oxide emissions from tropical forest soils exposed to elevated nitrogen input strongly interact with rainfall quantity and seasonality. *Biogeochemistry*, 118(1), 103–120. https://doi.org/10.1007/s10533-013-9908-3
- Deng, L., Huang, C., Kim, D.-G., Shangguan, Z., Wang, K., Song, X., & Peng, C. (2020). Soil GHG fluxes are altered by N deposition: New data indicate lower N stimulation of the N₂O flux and greater stimulation of the calculated C pools. *Global Change Biology*, 26(4), 2613–2629. https://doi.org/10.1111/gcb.14970
- Duan, P., Zhang, Q., Zhang, X., & Xiong, Z. (2019). Mechanisms of mitigating nitrous oxide emissions from vegetable soil varied with manure, biochar and nitrification inhibitors. Agricultural and Forest Meteorology, 278, 107672. https://doi.org/10.1016/j.agrformet.2019.107672
- Duan, P., Zhou, J., Feng, L., Jansen-Willems, A. B., & Xiong, Z. (2019). Pathways and controls of N₂O production in greenhouse vegetable production soils. *Biology and Fertility of Soils*, 55(3), 285–297. https://doi.org/10.1007/s00374-019-01348-9
- Enanga, E. M., Casson, N. J., Fairweather, T. A., & Creed, I. F. (2017). Nitrous oxide and dinitrogen: The missing flux in nitrogen budgets of forested catchments? *Environmental Science & Technology*, 51(11), 6036–6043. https://doi.org/10.1021/acs.est.6b03728
- Enanga, E. M., Creed, I. F., Casson, N. J., & Beall, F. D. (2016). Summer storms trigger soil N₂O efflux episodes in forested catchments. *Journal of Geophysical Research: Biogeosciences*, 121(1), 95–108. https://doi.org/10.1002/2015jg003027
- Fan, J., Xu, Y., Chen, Z., Xiao, J., Liu, D., Luo, J., et al. (2017). Sulfur deposition suppressed nitrogen-induced soil N₂O emission from a subtropical forestland in southeastern China. Agricultural and Forest Meteorology, 233, 163–170. https://doi.org/10.1016/j.agrformet.2016.11.017
- Fang, Y., Gundersen, P., Zhang, W., Zhou, G., Christiansen, J. R., Mo, J., et al. (2009). Soil-atmosphere exchange of N₂O, CO₂ and CH₄ along a slope of an evergreen broad-leaved forest in southern China. *Plant and Soil*, *319*(1), 37–48. https://doi.org/10.1007/s11104-008-9847-2

Acknowledgments

This work was funded by the National Natural Science Foundation of China (31,760,153; 32,001,175), Guangxi Bagui Scholarship Program to Dejun Li, National High–Level Talents Specical Support Program to Dejun Li, CAS Key Laboratory of Agro–ecological Processes in Subtropical Region (Y9550001), and Special Research Assistant Fellowship of Chinese Academy of Sciences to Pengpeng Duan (E0290502).

- Fu, Z., Niu, S., & Dukes, J. S. (2015). What have we learned from global change manipulative experiments in China? A meta-analysis. Scientific Reports, 5(1), 12344. https://doi.org/10.1038/srep12344
- Galloway, J. N., Dentener, F. J., Capone, D. G., Boyer, E. W., Howarth, R. W., Seitzinger, S. P., et al. (2004). Nitrogen cycles: Past, present, and future. *Biogeochemistry*, 70(2), 153–226. https://doi.org/10.1007/s10533-004-0370-0

Giles, M., Morley, N., Baggs, E. M., & Daniell, T. J. (2012). Soil nitrate reducing processes-drivers, mechanisms for spatial variation, and significance for nitrous oxide production. *Frontiers in Microbiology*, 3, 407. https://doi.org/10.3389/fmicb.2012.00407

Grave, R. A., Nicoloso, R. d. S., Cassol, P. C., da Silva, M. L. B., Mezzari, M. P., Aita, C., & Wuaden, C. R. (2018). Determining the effects of tillage and nitrogen sources on soil N₂O emission. *Soil and Tillage Research*, 175, 1–12. https://doi.org/10.1016/j.still.2017.08.011

Hallin, S., Philippot, L., Löffler, F. E., Sanford, R. A., & Jones, C. M. (2018). Genomics and ecology of novel N₂O-reducing microorganisms. *Trends in Microbiology*, 26(1), 43–55. https://doi.org/10.1016/j.tim.2017.07.003

Hall, S. J., & Matson, P. A. (1999). Nitrogen oxide emissions after nitrogen additions in tropical forests. Nature, 400(6740), 152–155. https://doi. org/10.1038/22094

Han, X., Shen, W., Zhang, J., & Müller, C. (2018). Microbial adaptation to long-term N supply prevents large responses in N dynamics and N losses of a subtropical forest. *The Science of the Total Environment*, 626, 1175–1187. https://doi.org/10.1016/j.scitotenv.2018.01.132

Han, X., Xu, C., Nie, Y., He, J., Wang, W., Deng, Q., & Shen, W. (2019). Seasonal variations in N₂O emissions in a subtropical forest with exogenous nitrogen enrichment are predominately influenced by the abundances of soil nitrifiers and denitrifiers. *Biogeosciences*, 124(11), 3635–3651, https://doi.org/10.1029/2019jg005477

Hao, T., Zhang, Y., Zhang, J., Müller, C., Li, K., Zhang, K., et al. (2020). Chronic nitrogen addition differentially affects gross nitrogen transformations in alpine and temperate grassland soils. Soil Biology and Biochemistry, 149, 107962. https://doi.org/10.1016/j.soilbio.2020.107962

Harter, J., Krause, H.-M., Schuettler, S., Ruser, R., Fromme, M., Scholten, T., et al. (2014). Linking N₂O emissions from biochar-amended soil to the structure and function of the N-cycling microbial community. *The ISME Journal*, 8(3), 660–674. https://doi.org/10.1038/ismej.2013.160

Hu, H., Chen, D., & He, J. (2015). Microbial regulation of terrestrial nitrous oxide formation: Understanding the biological pathways for prediction of emission rates. FEMS Microbiology Reviews, 39(5), 729–749. https://doi.org/10.1093/femsre/fuv021

- Jan-Michael, B., Qingxian, S., Yunjie, M., Borja, V.-P., Carlos, D.-F., Mark, J. M., & Barth, F. S. (2018). The pH dependency of N-converting enzymatic processes, pathways and microbes: Effect on net N₂O production. *Environmental Microbiology*, 20(5), 1623–1640. https://doi. org/10.1111/1462-2920.14063
- Jones, C. M., Graf, D. R., Bru, D., Philippot, L., & Hallin, S. (2013). The unaccounted yet abundant nitrous oxide-reducing microbial community: A potential nitrous oxide sink. *The ISME Journal*, 7(2), 417–426. https://doi.org/10.1038/ismej.2012.125

Koehler, B., Corre, M. D., Veldkamp, E., Wullaert, H., & Wright, S. J. (2009). Immediate and long-term nitrogen oxide emissions from tropical forest soils exposed to elevated nitrogen input. *Global Change Biology*, 15(8), 2049–2066. https://doi.org/10.1111/j.1365-2486.2008.01826.x

Levy-Booth, D. J., Prescott, C. E., & Grayston, S. J. (2014). Microbial functional genes involved in nitrogen fixation, nitrification and denitrification in forest ecosystems. Soil Biology and Biochemistry, 75, 11–25. https://doi.org/10.1016/j.soilbio.2014.03.021

Levy, P. E., Gray, A., Leeson, S. R., Gaiawyn, J., Kelly, M. P. C., Cooper, M. D. A., et al. (2011). Quantification of uncertainty in trace gas fluxes measured by the static chamber method. *European Journal of Soil Science*, 62(6), 811–821, https://doi.org/10.1111/j.1365-2389.2011.01403.x

Li, D., Liu, J., Chen, H., Zheng, L., & Wang, K. (2018). Soil gross nitrogen transformations in responses to land use conversion in a subtropical karst region. Journal of Environmental Management, 212, 1–7. https://doi.org/10.1016/j.jenvman.2018.01.084

- Li, D., Yang, Y., Chen, H., Xiao, K., Song, T., & Wang, K. (2017). Soil gross nitrogen transformations in typical karst and nonkarst forests, Southwest China. Journal of Geophysical Research: Biogeosciences, 122(11), 2831–2840. https://doi.org/10.1002/2017JG003850
- Liu, L., & Greaver, T. L. (2009). A review of nitrogen enrichment effects on three biogenic GHGs: The CO₂ sink may be largely offset by stimulated N₂O and CH₄ emission. *Ecology Letters*, 12(10), 1103–1117. https://doi.org/10.1111/j.1461-0248.2009.01351.x
- Long, A., Song, B., Fridey, K., & Silva, A. (2015). Detection and diversity of copper containing nitrite reductase genes (nirK) in prokaryotic and fungal communities of agricultural soils. FEMS Microbiology Ecology, 91(2), 1–9. https://doi.org/10.1093/femsec/fiu004
- Maharjan, B., & Venterea, R. T. (2013). Nitrite intensity explains N management effects on N₂O emissions in maize. Soil Biology and Biochemistry, 66, 229–238. https://doi.org/10.1016/j.soilbio.2013.07.015

Martinson, G. O., Corre, M. D., & Veldkamp, E. (2013). Responses of nitrous oxide fluxes and soil nitrogen cycling to nutrient additions in montane forests along an elevation gradient in southern Ecuador. *Biogeochemistry*, 112(1), 625–636. https://doi.org/10.1007/s10533-012-9753-9

Müller, A. K., Matson, A. L., Corre, M. D., & Veldkamp, E. (2015). Soil N₂O fluxes along an elevation gradient of tropical montane forests under experimental nitrogen and phosphorus addition. *Frontiers of Earth Science*, 3(66). https://doi.org/10.3389/feart.2015.00066

Oikawa, P. Y., Ge, C., Wang, J., Eberwein, J. R., Liang, L. L., Allsman, L. A., et al. (2015). Unusually high soil nitrogen oxide emissions influence air quality in a high-temperature agricultural region. *Nature Communications*, 6, 8753. https://doi.org/10.1038/ncomms9753

Prosser, J. I., Hink, L., Gubry-Rangin, C., & Nicol, G. W. (2020). Nitrous oxide production by ammonia oxidizers: Physiological diversity, niche differentiation and potential mitigation strategies. *Global Change Biology*, 26(1), 103–118. https://doi.org/10.1111/gcb.14877

Ravishankara, A. R., Daniel, J. S., & Portmann, R. W. (2009). Nitrous oxide (N₂O): The dominant ozone-depleting substance emitted in the 21st century. *Science*, 326(5949), 123–125. https://doi.org/10.1126/science.1176985

Redding, M. R., Shorten, P. R., Lewis, R., Pratt, C., Paungfoo-Lonhienne, C., & Hill, J. (2016). Soil N availability, rather than N deposition, controls indirect N₂O emissions. Soil Biology and Biochemistry, 95, 288–298. https://doi.org/10.1016/j.soilbio.2016.01.002

Rowlings, D. W., Grace, P. R., Kiese, R., & Weier, K. L. (2012). Environmental factors controlling temporal and spatial variability in the soil-atmosphere exchange of CO₂, CH₄ and N₂O from an Australian subtropical rainforest. *Global Change Biology*, 18(2), 726–738. https://doi. org/10.1111/j.1365-2486.2011.02563.x

- Shi, X., Hu, H.-W., Wang, J., He, J.-Z., Zheng, C., Wan, X., & Huang, Z. (2018). Niche separation of comammox Nitrospira and canonical ammonia oxidizers in an acidic subtropical forest soil under long-term nitrogen deposition. Soil Biology and Biochemistry, 126, 114–122. https:// doi.org/10.1016/j.soilbio.2018.09.004
- Song, L., Tian, P., Zhang, J., & Jin, G. (2017). Effects of three years of simulated nitrogen deposition on soil nitrogen dynamics and greenhouse gas emissions in a Korean pine plantation of northeast China. *The Science of the Total Environment*, 609, 1303–1311. https://doi.org/10.1016/j. scitotenv.2017.08.017
- Song, X., Peng, C., Ciais, P., Li, Q., Xiang, W., & Xiao, W. (2020). Nitrogen addition increased CO₂ uptake more than non-CO₂ greenhouse gases emissions in a Moso bamboo forest. *Science Advances*, 6(12), eaaw5790. https://doi.org/10.1126/sciadv.aaw5790
- Soper, F. M., Sullivan, B. W., Nasto, M. K., Osborne, B. B., Bru, D., Balzotti, C. S., et al. (2018). Remotely sensed canopy nitrogen correlates with nitrous oxide emissions in a lowland tropical rainforest. *Ecology*, 98(9), 2080–2089. https://doi.org/10.1002/ecy.2434
- Srikanthasamy, T., Leloup, J., N'Dri, A. B., Barot, S., Gervaix, J., Koné, A. W., et al. (2018). Contrasting effects of grasses and trees on microbial N-cycling in an African humid savanna. Soil Biology and Biochemistry, 117, 153–163. https://doi.org/10.1016/j.soilbio.2017.11.016



- Stewart, K. J., Grogan, P., Coxson, D. S., & Siciliano, S. D. (2014). Topography as a key factor driving atmospheric nitrogen exchanges in arctic terrestrial ecosystems. Soil Biology and Biochemistry, 70, 96–112. https://doi.org/10.1016/j.soilbio.2013.12.005
- Tang, Y., Zhang, X., Li, D., Wang, H., Chen, F., Fu, X., et al. (2016). Impacts of nitrogen and phosphorus additions on the abundance and community structure of ammonia oxidizers and denitrifying bacteria in Chinese fir plantations. Soil Biology and Biochemistry, 103, 284–293. https:// doi.org/10.1016/j.soilbio.2016.09.001
- Tian, D., & Niu, S. (2015). A global analysis of soil acidification caused by nitrogen addition. *Environmental Research Letters*, 10(2), 024019. https://doi.org/10.1088/1748-9326/10/2/024019
- Tian, H., Xu, X., Liu, M., Ren, W., Zhang, C., Chen, G., & Lu, C. (2010). Spatial and temporal patterns of CH₄ and N₂O fluxes in terrestrial ecosystems of North America during 1979-2008: Application of a global biogeochemistry model. *Biogeosciences*, 7(9), 2673–2694. https:// doi.org/10.5194/bg-7-2673-2010
- Tian, H., Yang, J., Xu, R., Lu, C., Canadell, J. G., Davidson, E. A., et al. (2019). Global soil nitrous oxide emissions since the preindustrial era estimated by an ensemble of terrestrial biosphere models: Magnitude, attribution, and uncertainty. *Global Change Biology*, 25(2), 640–659. https://doi.org/10.1111/gcb.14514
- Wang, Z., Li, D., Zheng, M., Chen, H., Sun, X., & Wang, K. (2019). Topography modulates effects of nitrogen deposition on asymbiotic N₂ fixation in soil but not litter or moss in a secondary karst forest. *Journal of Geophysical Research: Biogeosciences*, 124(10), 3015–3023. https:// doi.org/10.1029/2019jg005291
- Wieder, W. R., Cleveland, C. C., & Townsend, A. R. (2011). Throughfall exclusion and leaf litter addition drive higher rates of soil nitrous oxide emissions from a lowland wet tropical forest. *Global Change Biology*, 17(10), 3195–3207. https://doi.org/10.1111/j.1365-2486.2011.02426.x
- Wrage-Mönnig, N., Horn, M. A., Well, R., Müller, C., Velthof, G., & Oenema, O. (2018). The role of nitrifier denitrification in the production of nitrous oxide revisited. Soil Biology and Biochemistry, 123, 3–16. https://doi.org/10.1016/j.soilbio.2018.03.020
- Xie, D., Si, G., Zhang, T., Mulder, J., & Duan, L. (2018). Nitrogen deposition increases N₂O emission from an N-saturated subtropical forest in southwest China. *Environmental Pollution*, 243, 1818–1824. https://doi.org/10.1016/j.envpol.2018.09.113
- Yin, C., Fan, F., Song, A., Fan, X., Ding, H., Ran, W., et al. (2017). The response patterns of community traits of N₂O emission-related functional guilds to temperature across different arable soils under inorganic fertilization. Soil Biology and Biochemistry, 108, 65–77. https://doi. org/10.1016/j.soilbio.2017.01.022
- Yu, L., Mulder, J., Zhu, J., Zhang, X., Wang, Z., & Dörsch, P. (2019). Denitrification as a major regional nitrogen sink in subtropical forest catchments: Evidence from multi-site dual nitrate isotopes. *Global Change Biology*, 25(5), 1765–1778. https://doi.org/10.1111/gcb.14596
- Yu, L., Zhu, J., Ji, H., Bai, X., Lin, Y., & Zhang, Y. (2021). Topography-related controls on N₂O emission and CH₄ uptake in a tropical rainforest catchment. *The Science of the Total Environment*, 775, 145616. https://doi.org/10.1016/j.scitotenv.2021.145616
- Zhang, K., Zhu, Q., Liu, J., Wang, M., Zhou, X., Li, M., et al. (2019). Spatial and temporal variations of N₂O emissions from global forest and grassland ecosystems. Agricultural and Forest Meteorology, 266–267, 129–139. https://doi.org/10.1016/j.agrformet.2018.12.011
- Zheng, L., Chen, H., Wang, Y., Mao, Q., Zheng, M., Su, Y., et al. (2020). Responses of soil microbial resource limitation to multiple fertilization strategies. Soil and Tillage Research, 196, 104474. https://doi.org/10.1016/j.still.2019.104474
- Zheng, M. H., Zhang, T., Li, L., Zhu, W. X., Zhang, W., & Mo, J. M. (2016). Effects of nitrogen and phosphorus additions on nitrous oxide emission in a nitrogen-rich and two nitrogen-limited tropical forests. *Biogeosciences*, 13, 3503–3517. https://doi.org/10.5194/bg-13-3503-2016
- Zhu, J., He, N., Wang, Q., Yuan, G., Wen, D., Yu, G., & Jia, Y. (2015). The composition, spatial patterns, and influencing factors of atmospheric wet nitrogen deposition in Chinese terrestrial ecosystems. *The Science of the Total Environment*, 511, 777–785. https://doi.org/10.1016/j. scitotenv.2014.12.038